\label{Docs_Creating_From_Template}

Several templates are available for you
to fill out and add to your course.

To add a templated document to your course, click on 
\textbf{Special Documents} at the buttom of the \textbf{Course Editor} screen.
Select the templated
\includegraphics[width=0.03\paperwidth]{pref}
\textbf{Syllabus}, 
\includegraphics[width=0.03\paperwidth]{pref}
\textbf{Simple Course Page}, 
\includegraphics[width=0.03\paperwidth]{pref}
\textbf{Discussion Board}, or 
\textbf{My Personal Information Page}. 
Clicking on 
\includegraphics[width=0.03\paperwidth]{pref}
\textbf{Simple Course Page} and 
\includegraphics[width=0.03\paperwidth]{pref}
\textbf{Discussion Board} will pop up a window asking you to give a title (if necessary). Once you've
titled the page, it will be added to the end of the current folder of the
\textbf{Course Editor} screen.

NOTE: Your new resources will be visible the next time you log in, or by clicking the \fbox{re-intializing course} button on the \textbf{Course Editor} screen. 
